

Temi di discussione

(Working Papers)

The impact of sovereign tensions on bank lending: identifying the channels at work

by Fabiana Sabatini





Temi di discussione

(Working Papers)

The impact of sovereign tensions on bank lending: identifying the channels at work

by Fabiana Sabatini

Number 1397 - December 2022

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Lorenzo Braccini, Luca Citino, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Dario Ruzzi, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

THE IMPACT OF SOVEREIGN TENSIONS ON BANK LENDING: IDENTIFYING THE CHANNELS AT WORK

by Fabiana Sabatini*

Abstract

Banks' holdings of sovereign bonds are an important component of the multifaceted bank-sovereign nexus. This paper exploits the unexpected increase in sovereign yields in Italy in May 2018 to quantify the impact of a drop in value of banks' government bond portfolios on their supply of loans (direct channels). It disentangles the effect stemming from the worsening in banks' capitalization (balance sheet channel) from that associated with a reduced ability to raise funds using government bond holdings as collateral (liquidity channel). Results show that banks with large government bond portfolios reduced their lending more; evidence indicates that this is a consequence of the balance-sheet channel. The liquidity channel was not activated, partly thanks to the ample availability of Eurosystem funds held by banks. I then control for the channels at work for the banking system as a whole, regardless of government bond holdings (indirect channels), and find that the generalized increase in the cost of funding for banks (cost of funding channel) has a negative impact on bank lending.

JEL Classification: G01, G21, E5, E51. Keywords: credit supply, banks' sovereign holdings, bank lending channel. DOI: 10.32057/0.TD.2022.1397

Contents

1. Introduction	5
2. Related literature	9
3. Data	10
3.1 The classification of bonds under the IFRS 9	11
3.2 Variables used in the regression	12
4. Empirical strategy	13
4.1 Identification issues	
4.2 Model	15
5. Results	17
6. Extensions	18
6.1 Taking into account the indirect channels of sovereign tensions	18
6.2 Private markets and Eurosystem operations	20
7. Concluding remarks	22
References	24
Tables and figures	27
Appendix	

^{*} Bank of Italy, Economic Outlook and Monetary Policy Directorate.

1 Introduction¹

A deterioration of sovereign creditworthiness may negatively affect the supply of bank lending in several ways.² Two important channels are directly related to the amount of sovereign bonds that banks hold on their balance sheets (*direct channels*). The first is activated by the impact on bank capitalization exerted by a reduction in the value of sovereign bonds: indeed, the unrealized losses on sovereign portfolio at market values dent capital position (bank balance-sheet channel; see Albertazzi et al. 2014). The second is triggered by the reduction in the amount of funds that banks can raise in collateralized wholesale markets or through Eurosystem operations, since government bonds are largely used as collateral (liquidity channel; see Angelini et al. 2014).

In addition to these two mechanisms, sovereign risk may deteriorate bank's funding conditions - and in turn impair the supply of loans to the private sector - via a number of *indirect channels*, which are at work for all banks and whose activation and intensity do not depend on the amount of sovereign bonds held by a given institution. Among these: an increase in sovereign yields determines a rise in the cost of bank funding as country risk normally acts as a floor for the cost of issuing bonds (cost of funding channel, see BIS 2011); the reduction in the benefits of implicit and -if present- explicit government guarantees can increase banks risk premia (government guarantee channel; see Mäkinen et al. 2020 and Correa et al. 2014); sovereign downgrades often lead to downgrades of banking institutions as sovereign ratings are normally a ceiling for the private borrowers (rating downgrade channel, see Adelino and Ferreira 2016).

Finally, the reduction in lending supply may be due to other factors rather than the deterioration in credit institutions' conditions; for example, a large strand of the literature documents that public debt portfolios held by banks expanded after the sovereign debt crisis, reflecting either government pressure or risk-shifting motives (see Popov and Van Horen 2014 and Becker and Ivashina 2018). Whatever the reason behind it, the increase in sovereign bonds holdings may be conducted at the expense of loan granting, causing a decline in credit supply (crowding-out effect of loans).

The main objective of this paper is to provide a quantitative estimation of the balancesheet *versus* the liquidity channel within the direct channels; the focus is on credit supply granted by Italian banks between June 2018 and December 2018 - i.e., in the six months following the increase observed in Italian sovereign yields in May 2018. While the literature has established a significant positive relation between the overall amount of government bonds held by banks and the reduction in credit supply following an increase in sovereign risk (Popov and Van Horen 2014, Bottero et al. 2020), a quantitative assessment of the

¹I am grateful to Fabiano Schivardi for guidance and support. I would like to thank Andrew Ellul, Filippo De Marco, Andrea Polo, Alessandro Secchi, Giuseppe Ferrero, Federico Maria Signoretti, Margherita Bottero, Massimiliano Affinito, Maria Sole Pagliari, Stefano Schiaffi, Francesco Palazzo and Giovanni Soggia for useful comments and suggestions, as well as seminar participants at the Bank of Italy, Luiss and 2022 EEA Meeting. The views expressed in this paper do not necessarily reflect those of Bank of Italy. All errors are the responsibility of the author.

 $^{^{2}}$ For a more detailed description of the cannels see BIS (2011)

impact of each of the two aforementioned channels is still lacking - to the best of my knowledge. Given that sovereign tensions can affect, via direct channels, both lending and real variables such as firms' investments and employment (see Altavilla et al. 2015, Bottero et al. 2020 and De Marco 2018), a deep understanding of the mechanisms at work in their transmission provides policymakers with useful information for the regulatory treatment of government bonds in banks' portfolios.

The key factor that allows to disentangle the two mechanisms is that the changes in the value of a government bond are treated differently depending on the portfolio classification of the asset. In particular, according to the International Financial Reporting Standard 9 (IFRS 9), a financial asset can be classified into the following portfolios: (i) *held to collect*, (ii) *fair value through other comprehensive income* and (iii) *fair value through profit and loss*. Financial assets not intended to be sold can be allocated in portfolio (i) and are held at the amortised cost: this means that changes in their value will not affect capital position and the balance-sheet channel won't be activated for these bonds; assets classified under (ii) and (iii) are instead held at market values, implying that changes in their price impact capital position. On the contrary, in case of a price variation, the liquidity channel is triggered for all bonds as their valuation as collateral is independent of the portfolio under which they are allocated.

Summarizing, the amount of securities held at market value will affect the intensity of both channels, while the amount of securities at amortised cost will determine the intensity of the liquidity channel only. A direct consequence is that the growth of loans after the outbreak of sovereign tensions can be associated with the pre-shock share of sovereign debt under each portfolio in order to disentangle the impact of each channel on lending policies. Specifically, when included in a regression having loan growth as the dependent variable, incidence on total assets of (i) sovereign exposures and (ii) sovereign exposures held in the fair value portfolios will provide an estimate of the relevance of, respectively, the liquidity channel and the bank balance-sheet channel.

A number of conditions are required for these estimates to measure the impact of sovereign tensions on lending supply. First, the shock under study should be exogenous to the banking system - i.e., it should not be caused by a deterioration in banks' health. To this end, I exploit the episode of sharp rise in the Italian sovereign yields observed in the Spring of 2018 and connected with the sudden increase in the political uncertainty. In the second half of May, after the failure of several attempts to form a new government, yields on sovereign bonds increased by around 200, 175 and 110 basis points on the two-, five- and ten-year horizons respectively. Given that this rise was unexpected and mainly reflected the high uncertainty over the formation of the new government, the shock can be regarded as exogenous, making this event an ideal setting to study the impact of sovereign tensions on the supply of credit. Its exogenous nature is confirmed by the fact that, in the period preceding the rise in sovereign yields, indicators for the Italian banking system were pointing to sound and improving conditions: during 2017 and in the first months of 2018 banks' profitability and capitalization strengthened markedly, while non-performing loans ratio declined to the pre-sovereign crisis levels.

The shock exploited in this paper presents two advantages with respect to the one observed during the sovereign debt crisis and largely used in the literature to estimate the impact of sovereign tensions on credit supply. A first advantage is that the rise in government yields in 2018 was unanticipated, as it was connected to the sudden rise in the uncertainty of the political background. A second one is that the shock observed in 2018 did not involve other countries apart from Italy, so I am able to properly estimate the effects of sovereign tensions originated in a specific country on its bank credit supply.

A second condition required to the aim of this study is that loan supply must be isolated from loan demand: in order to do this, I follow the methodology pioneered by Khwaja and Mian (2008) by means of a granular dataset comprising 1,047,378 observations at bank-firm level: the proper identification of credit supply is possible by adding in the regressions firmtime fixed effects that control for observed and unobserved firm heterogeneity - and thus for the demand of credit. Third, banks' investment and lending policies may be commonly influenced by non-observables characteristics, i.e. the bank business model, potentially raising an issue of identification. This identification issue is dealt with by including bank fixed effects: controlling for time-invariant unobserved characteristics of banks on such a short horizon, bank business model is reasonably controlled for because this feature does not change frequently over time.

The results show that, as a consequence of the outburst of sovereign tensions in the Spring of 2018, government bonds portfolios are negatively associated with the growth of credit supply. In addition the drop in credit supply was triggered by the *balance-sheet channel*, while I find no evidence of an activation of the *liquidity channel*. The effect of this channel is remarkably smaller than that found by similar studies on 2021-2011, suggesting that the sovereign debt crisis hit the banking system more severely than the shock under study here. Specifically, a one-standard deviation increase in the share of government bonds at fair value is associated with a reduction in the growth rate of credit to a given firm slightly lower than one percentage points; for comparison, De Marco (2018) finds that a one standard deviation increase in the market-to-market sovereign exposure is associated with a drop in the growth of credit of almost 6 percentage points, while, after recasting their effects in percentage points, a one standard deviation increase in the sovereign to assets variable implies a reduction of around 4 percentage points in Bottero et al. (2020). After estimating the baseline model, I extend the analysis along two dimensions. First, I add variables measuring the impact of the indirect channels, which might have also affected lending supply during the horizon considered. In particular, I include proxies for (i) the cost of funding channel, (ii) the (implicit or explicit) public guarantees channel and (iii) the crowding-out effect of loans.³ Results on the direct channels hold. At the same time, regressions indicate that also the cost of funding channel was also an important driver of

 $^{^{3}}$ I deliberately disregard the rating channel, which is unlikely to have had any role in the period considered: in 2018 only Moody's - among the most important credit rating agencies - downgraded the Italian sovereign rating, only by one notch and within the investment-grade class (to Baa3).

the reduction in lending supply. This result is consistent with Bofondi et al. (2017), according to whom during the most acute phase of sovereign debt crisis the generalized rise in the cost of funding was the main driver for the much stronger reduction in lending by Italian domestic banks with respect to that carried-out by foreign branches. As about the economic magnitude, an increase of a one standard deviation in the proxy for the funding channel is associated with a 1.1 percentage points reduction in the growth rate of loan supply, comparable to that associated with the balance-sheet channel - and smaller than the one found by Bofondi et al. (2017) (3 percentage points).

Second, I check whether the abundant liquidity provided by the ECB to the euro area banking system may explain the finding that the liquidity channel did not activate following the 2018 sovereign shock. This hypothesis would reconcile the result of the baseline with those found by studies on the sovereign debt crisis, according to which the strains on liquidity wholesale markets for banks was arguably the main factor behind the tightening of loan supply conditions since the end of 2011. In order to do so, for each bank I compute the share of total funds raised on private collateralized markets (with either euro area private banks or central counterparties) over the total collateralized funding (which comprises also the liquidity raised via Eurosystem operations). Then, I add the interaction between this variable and the government bonds holdings in the baseline equation: the coefficient on the interaction term is negative and significant, suggesting that larger recourse to the market (as compared with the Eurosystem operations) is associated with an activation of the liquidity channel. As for the magnitude, the results indicate that for the same borrower, the growth of credit from a hypothetical bank raising funds on the market only was 0.3 percentage points lower than that of a hypothetical bank recurring only to the Eurosystem operations.

Summing up, the analysis suggests that the size of banks' sovereign portfolios matters in the transmission of sovereign shocks and affects credit supply via an adverse impact on both capitalization and funding conditions on wholesale markets. At the same time, when controlling for the other indirect channels of the multifaceted bank-sovereign nexus, the contribution of banks' direct exposure is not the most important mechanism, as the economic significance associated with the cost of funding channel is slightly larger.

The paper contributes to the empirical literature is several ways. First, it provides a novel quantification of the effects of a rise in sovereign yields on bank supply of credit by distinguishing, within the direct channels, the bank-balance sheet and liquidity channels. Second, this is the first work, to the best of my knowledge, to compare the direct and indirect channels through which a shock to government financing costs can propagate to the availability of funds to the economy. Finally, the paper investigates how the stance of monetary policy can affect the propagation of sovereign tensions to bank supply of credit, and provides evidence that the non-activation of the liquidity channel is due to the large recourse of credit intermediaries to Eurosystem funds instead of private markets ones. As an important *caveat*, this analysis does not take into account that the direct holdings of

As an important *caveat*, this analysis does not take into account that the direct holdings of government bonds by credit institutions has also important beneficial effects for financial

and, in turn, macroeconomic stability. This is for example related to banks' tendency to act as contrarian investors in sovereign bond markets during episodes of financial tensions (Lanotte et al. 2016).

The rest of the paper is structured as follows: Section 2 reviews the related literature; Section 3 describes the data, Section 4 addresses the identification issues and describes the empirical strategy. Section 5 presents the results of the baseline and some robustness checks, while Section 6 presents the extensions. Section 7 briefly discusses the evidence presented and concludes.

2 Related literature

This paper relates to two strands of literature. First, it is connected to the studies analysing the relationship between banks' balance-sheet strength and credit availability (the so called bank lending channel; see Bernanke and Gertler 1995). A number of studies have documented how lending supply is crucially affected by the degree of bank capitalization. In their seminal work, Bernanke and Lown (1991) document that the credit crunch observed in the United States in the early nineties may be partly attributed to the shortage of capital suffered by credit institutions. Other studies have also highlighted the role of funding conditions and liquidity of bank balance-sheets in determining the effects of monetary policy and other shocks on credit supply (Kashyap and Stein, 1995; Stein and Kashyap, 2000). In recent years, after the pioneering work by Khwaja and Mian (2008), the use of large bank-firm matched datasets to study adverse shocks to credit supply has become more and more frequent. Among these, Jimenèz et al. (2012) find that the tightening in credit supply following a worsening in macroeconomic conditions, as higher interest rates or lower GDP, is stronger for banks with weaker capital and liquidity conditions; Jiménez et al. (2014), Schivardi et al. (2017), Acharya et al. (2019) find that low capitalized banks supply more credit to riskier firms than better capitalized banks. Other works show that adverse shocks on bank funding are transmitted to credit supply to non-financial sector, either as a reduction in credit granted (Puri et al. 2011) or as a reallocation of credit portfolio toward low-risk firms (Olivier et al. 2020). This work contributes to this strand of literature by investigating the effects on credit supply of a sudden deterioration in banks' conditions; importantly, and differently from previous works, the methodology proposed aims at disentangling the effects of the shock on the liquidity position from those of the shock on the capital position.

The second strand of literature to which this paper contributes concerns the impact of sovereign tensions on bank credit supply. In a precursor work, Arteta and Hale (2008) study the effect of sovereign debt crises in emerging markets and find that these are associated with a reduction of foreign credit to domestic firms. Since then, a large number of works have focused on the impact of the European sovereign debt crisis of 2011-2012. Bofondi et al. (2017) find that during the sovereign debt crisis domestic banks in Italy reduced credit supply by more than foreign banks, as a consequence of a rise in their

cost of funding. Some studies focused on the impact of the direct holdings of government securities on credit supply: Becker and Ivashina (2018), using a dataset covering the period 1995-2015 identify the set of firms with a positive demand for funds; for these firms, they document that whenever the potential lender was a bank with a high exposure to sovereign domestic debt, then the firm was more likely to increase the debt issuance rather than bank credit: this result is interpreted as a consequence of the reduced credit supply by the banks most exposed to sovereign debt; Popov and Van Horen (2014), using a dataset on European syndicated loans, document that banks exposed to stressed euro-area public debt reduced lending to firms by more than less exposed banks; Bottero et al. (2020) use firm-bank matched data from the Italian Credit Register and find that during periods of distress on sovereign debt markets higher direct exposures are associated with a stronger reduction in credit supply. De Marco (2018) is the closest work to this paper, as he analyses the mechanism whereby sovereign stress propagates to credit supply via public debt holdings: he finds that the negative effect on credit supply is stronger for banks with a higher share of short-term funding while the level of capitalization does not matter. This result is interpreted as evidence that in the period considered a cost of funding channel was at work while a bank balance-sheet channel was not. Differently from the previous studies, this paper proposes a method to disentangle the two main mechanisms connected with the direct public bonds holdings (the bank balance-sheet and the liquidity channels), based on the accounting classification of bonds.

3 Data

I use a unique dataset at bank-firm level obtained by merging the information from the Italian Credit Register (CR) and the Supervisory Reports statistics. The CR contains information at loan level on credit granted in Italy by credit institutions to non-financial borrowers of outstanding amount above 30,000 euros (250 euros for bad loans). The dataset distinguishes between three types of loans: revolving credit lines, term loans, loans backed by accounting receivables. Observations are collected at a quarterly frequency and I focus on credit granted, which is less affected by firms' decisions - as compared to drawn credit - and thus is better suited to capture the dynamics of loan supply (for more details on the dependent variable, see Section 3.2). Information on bank balance sheets are obtained by the Supervisor Reports statistics. Banks are required by law to report information on balance sheet quantities on a monthly basis (or quarterly for certain items). As it will be explained in detail below, I collect information on banks' capitalization, sovereign exposure, liquidity and funding structure, profitability and total assets. I collect data at four different dates in 2018: March, June, September and December. Banking information before January 2018 cannot be used because the implementation of the International Financial Reporting Standard 9 (IFRS 9) caused statistical discontinuity in several bank characteristics, notably in the accounting classification of sovereign bond portfolios (see Section 3.1 for more details). The dataset includes all firms borrowing from at least 2 banks in each of the 4 periods; overall, the sample comprises 141,749 firms borrowing from 181 banks, for a total of 469,046 bank-firm relationships.

3.1 The classification of bonds under the IFRS 9

The International Financial Reporting Standard 9 (IFRS 9) came into force in January 2018 and replaced the earlier International Accounting Standard 39 (IAS 39). According to the new standard, a financial asset can be allocated in one of the following portfolios: (i) *Held to collect*, (ii) *Fair value through other comprehensive income*, or (iii) *Fair value through profit and loss*. Assets under portfolio (i) are valued at the amortized cost, while assets under portfolios (ii) and (iii) are valued at fair value.

IFRS 9 aims at reducing banks' discretion in the allocation of financial assets in different portfolios. For this purpose, it requires to carry-out two tests: (i) the *solely payments of principal and interests test* (SPPI test), which requires that the asset gives rise to cash flows that are solely payments of principal and interests on the principal amount outstanding; (ii) the *business models test*, which evaluates the purpose for which the asset is held. Government bonds always satisfy the SPPI test, so their allocation reflects only the result of the business model test. Specifically, if the bank holds the government bond with the only purpose of collecting cash flows over the life on the instrument, then the government bond will be assigned to the *Held to collect* portfolio. Instead, if the bank pursues both the collection of contractual cash flows and the sale of financial assets, the asset will be recorded under the *Fair value through other comprehensive income* portfolio. Finally, in case of other businesses models (i.e., trading models), the security will be recorded under the *Fair value profit and loss* portfolio.

It is important to stress the following points. First, the result of the business model test, that identifies the first allocation of the asset, is conducted by taking into account objective elements of the investment and not a simple assertion by the bank: the entity must consider all relevant evidence that is available at the date of the assessment (as for instance, it is necessary to consider the frequency, value and timing of sales in prior periods, the reasons for those sales and expectations about future sales activity⁴). This prevents that banks use discretion in the allocation of bonds to portfolios. Second, subsequent movements of assets across portfolios are very difficult, as reclassification into other categories is possible only by changing the business model behind the assets.⁵

Under the assumption that the direct channels comprise only the balance sheet and the liquidity channel, the classification of government bonds held by banks provides a crucial information for disentangling the relative role of each direct channel, which is the main contribution of this paper. The intuition behind this approach is that for bonds classified at amortized cost only the liquidity channel is activated: therefore, the impact on credit supply associated with the direct sovereign exposure for a hypothetical bank holding bonds

 $^{^4 \}mathrm{See}$ Regulation (EU) 2016/2067 of 22 November 2016.

⁵Reclassification of assets up to a 10 per cent of the value without a change in the business model is also possible, but only under strict conditions and limitations, ibidem.

only under portfolio (i) estimates the liquidity channel. Similarly, for a bank that allocates government bonds under portfolios (ii) and (iii), the development of credit granted associated with its sovereign exposure will be the sum of both the bank-balance sheet and the liquidity channel. Comparing the growth of credit to the same borrower by two banks, differing only in the allocation of government bonds in each portfolio, allows estimating the magnitude of each channel at work.

Information on portfolios is reported on a monthly basis and I can compute the relative share of government bonds held at amortised cost or at market values by each bank at every point of time. Figure 1 graphically reports the incidence of government bonds over total assets and government bonds held at fair value over total assets by significant and less significant institutions.

3.2 Variables used in the regression

Table 1 reports names and descriptions for the variables used in the regressions and Table 2 reports the summary statistics before the shock, in March 2018 (panel A) and after the shock, for the second half of 2018 (panel B).

The dependent variable is the growth of credit at the bank-firm level on a 3-month horizon; in line with the literature, the variable is built using credit granted, as opposed to drawn credit, as the variations in the latter depend also on firms' decisions and liquidity needs - and therefore might reflect demand side factors. Credit supply fell in the second half of 2018: average and median lending dynamics are -0.9 and -0.1 per cent respectively, from -0.6 and 0.0 observed in the pre-shock period.

The key regressors are the ratio of Italian government bonds over total assets (GovBonds) and Italian government bonds held at fair value over total assets (FVGovBonds): as showed in the next section, they estimate respectively the liquidity and the balance sheet channel. Domestic public bonds portfolio is an important fraction of banks' assets, representing around 10.3 per cent of total assets for the average credit institution in the pre-shock period; the share of public bonds held at fair value is lower and equal to 6 per cent in the pre-shock period. In the post-shock period the share over total assets of total public bonds increased to 11.2 per cent while that of public bonds held at fair value slightly diminished to 5.6 per cent. The difference in the dynamics of FVGovBonds and GovBonds in the post-shock period may partly reflect the different impact that a soaring in government bonds yields has on these variables: indeed, a rise in sovereign yields reduces the value of government bonds held at fair value but leaves unchanged the value of those held at amortised cost. To take into account the mechanical reduction of FVGovBonds with respect to GovBonds in the post-shock period, these variables are measured at the pre-shock period only (March 2018) and are time-invariant: therefore, only the interaction with the dummy *Post* is estimated because *FVGovBonds* and *GovBonds* are absorbed by bank fixed effects. Figure 1 shows that both *GovBonds* and *FVGovBonds* are differently distributed across bank type; the share of government bonds over total assets held by smaller banks (less significant institutions) is higher than 18 percentage points while for significant institutions this indicator is around 6 per cent. Less significant institutions also display a higher incidence of domestic bonds held at fair value (around 9 per cent), while for significant banks this incidence is around 4 per cent. The differences of the distributions across banks might reflect that both the choices on the dimension of public portfolio and on the allocation of bonds under different portfolios may be associated with different bank characteristics, i.e. business models; in order to address this issue, it is important to include bank fixed effects that control for the time-constant features of credit institutions.

Other time variant bank characteristics may be associated with lending policies and the allocation of bonds over different portfolios. To tackle this issue and in line with the literature, I add in each regression the following standard bank variables: *capital ratio* (the ratio of regulatory capital to risk weighted assets), *size* (log of total assets), *ROA* (bank profit to total assets), *liquidity ratio* (the ratio of cash plus non-domestic government securities to total assets), *interbank funding ratio* (wholesale deposits to total assets). To take into account that in the post-shock period the capitalization of banks with a higher incidence of government bonds held at fair value reduces mechanically more than that of banks with a lower incidence, I measure *capital ratio* at the pre-shock period only and, as for *GovBonds* and *FVGovBonds*, only the interaction with *Post* is estimated.

Finally, to control for the features of a single lending relation, the following variables at bank-firm time level are also included: share of $credit_{ijt} = \frac{credit_{ijt}}{\sum_i credit_{ijt}}$ is the ratio between the size of the credit granted at bank-firm level over the total bank credit that the firm obtains; it measures how much the firm is dependent from the bank and, when equal to 1, the bank is the only source of bank credit for the firm. $bind_{ijt} = \frac{granted_{ijt} \cdot drawn_{ijt}}{granted_{ijt}}$ is computed as the difference between granted and drawn credit over credit granted: when zero, it signals that if the firm needs a new loan then it must apply for it.

4 Empirical strategy

4.1 Identification issues

Identifying the direct channels through which sovereign tensions impact on credit supply to firms poses relevant identification issues. In this section I will argue that the empirical strategy chosen is well-suited to address them.

A first identification issue is that the increase in Italian yields must be exogenous with respect to the health of Italian banking system and in particular it must not be caused by a deterioration in banks' conditions. The paper focuses on the sudden increase in the Italian sovereign yields recorded in the second half of May 2018 and observed against a political background that became increasingly unstable. In particular, when several attempts to form a new government failed in the second half of May, the uncertainty over the formation of the government suddenly rose. The increase in the uncertainty impacted on the Italian sovereign market: in less than two weeks, between the 18th and the 30rd of

May, yields on government bonds increased by around 200, 175 and 110 basis points on the two-, five- and ten-year horizons respectively (Fig.2). For the identification strategy it is important to stress that when the rise in sovereign yields occurred, the Italian banking system was sound and indicators were pointing to improving conditions: during 2017 and in the first months of 2018 banks' profitability grew markedly (for significant groups, annualized return on equity rose to 8.4 per cent in the first quarter of 2018 from 5.1 per cent in the first three months of 2017⁶); non-performing loans ratio declined to the pre-sovereign crisis levels; banks' capital ratio strengthened and the gap with other European banks narrowed.⁷ Summarizing, the shock arose from the sharp rise in political instability and did not originate from a deterioration in the banking conditions system, and can therefore be regarded as exogenous to the banking system.

The event exploited in this paper is an ideal setting and presents two advantages with respect to previous studies analysing the impact of sovereign tensions on credit supply by exploiting the sovereign debt crisis. One is that the rise in government yields in 2018 was unanticipated, as it was connected to the sudden rise in the uncertainty of the political background; on the contrary, international sovereign debt markets were under stress since the Greek bailout in 2010, well before yields on Italian sovereign securities markedly rose in July 2011; when the Italian government securities yields abruptly rose, this increase partly incorporated fears over the conditions of the Italian banking system: this poses serious problems in estimating the causal effect of sovereign tensions on the banking credit supply. Another advantage is that the development in government debt markets observed in 2018 did not involve other countries apart from Italy: therefore, I can study the effects of sovereign tensions originated in a country on the credit supply of banks operating in that specific country, ruling out the possibility that the estimates are affected by macro-financial shocks originated outside the national borders and transmitted to domestic financial institutions.⁸

A second identification issue is that the supply of credit needs to be properly isolated from the demand of credit. The whole dynamic of credit reflects firms' demand for funds and is, in turn, affected by macroeconomic conditions; in particular, in a context of high political uncertainty, non-financial corporations might temporally reduce the demand of funds for investment purposes. I deal with these aspects by following the methodology pioneered by Khwaja and Mian (2008). Firm-time fixed effects are included in the regressions and all time-varying unobserved heterogeneity at firm level is controlled for: thus, I control for any shock originated from firms. This approach allows for unbiased estimates as long as the demand shocks are not bank-specific: in other words, in the post-shock period firms' demand should not differ according to the amount of government securities held by banks

 $^{^6 \}mathrm{See}$ Economic Bulletin, 2018, 3.

 $^{^7\}mathrm{See}$ Economic Bulletin, 2018, 3 and Financial Stability Report, 2019, 1.

⁸A large literature investigates the international transmission of financial shocks: Peek and Rosengren (1997) focus on cross-borders lending; Schnabl (2012), Baskaya et al. (2017) focus on the propagation via international funding of domestic banks; finally, Cetorelli and Goldberg (2011), Peek and Rosengren (2000) investigate the role played by global banks.

or to the relative share of each accounting portfolio, which is a plausible assumption.

A third issue for identification is that banks' lending policies and the allocation of bonds under different portfolios might be correlated and explained by bank characteristics. This aspect is addressed in two ways. First, in order to control for time-varying observable heterogeneity among credit institutions, the regression includes bank variables that, according to the literature, influence the amount of government bonds held by banks⁹ (Section 3.2 provide details on the variables used in the regressions). However, the correlation between lending policies and allocations of public bonds over different portfolios might still be explained by unobservable factors, namely the bank business model; in order to deal with this aspect, the inclusion of bank fixed effects would control for the non-observable constant heterogeneity at bank level and, on a short span of time, also for bank business model (since this is very persistent over time).

The final dataset covers the whole 2018 and comprises 1 pre-shock observation and 2 post-shock observations, and bank fixed effects can be included. Ideally, I would have included another pre-shock observation but going backward is not possible as the entry into force of the International Financial Reporting Standard 9 (IFRS 9) in January 2018 caused statistical discontinuities in the classification of government bonds among different accounting portfolios.¹⁰ Another issue to deal with is the differences in the measurement for share of government bonds under the different portfolios in the post-shock period: indeed, the increase in yields on sovereign securities implies a mechanical reduction of the exposure of sovereign bonds held at fair value with respect to the exposures of sovereign bonds at amortised cost. This problem is solved by taking the sovereign exposures in the pre-shock period only.¹¹

4.2 Model

In order to identify the direct channels through which sovereign tensions affect the total amount of bank credit supply, the paper estimates a model at bank-firm level; the sample period comprises 4 quarterly data over 2018 and the shock is at Q2. Bank variables are included in every regression in order to control for bank heterogeneity; all independent variables are lagged because decision, approval and grant of credit require sometimes to be taken by the bank. Specifically, I am assuming that the rise in government yields that occurred at the end of May will take some time to affect credit supply, and its effect is not exerted already in June. Bank fixed effects control for time-invariant observed and unobserved characteristics, among which also the business model. The variables exploited to

⁹See, for instance, Bottero et al. (2020) or Gennaioli et al. (2018).

¹⁰In January 2018 the IFRS 9 replaced the earlier International Accounting Standard 39 (IAS 39); the coming into force of the new principle was followed by a strong increase at the aggregate level of the share of Government bonds held in the portfolios measured at amortised cost; this likely reflected that it was no longer possible for less significant institutions (LSI) to neutralize the capital gain or capital losses of the financial assets classified as 'Available for sale' as under IAS 39 (i.e. to opt for the so called *prudential filter*. More details can be found at: 5th update of Circular No. 263, Bank of Italy (Italian only).

 $^{^{11}\}mathrm{See}$ the next subsection for more details.

identify the channels are the ratio over total assets of total government bonds, $GovBonds_{i,t}$, and the ratio over total assets of total government bonds held at fair value, $FVGovBonds_{i,t}$. The model to be estimated is:

$$\Delta b_{ij,t+1} = \alpha \text{GovBonds}_{i,pre} \times \text{Post} + \beta \text{FVGovBonds}_{i,pre} \times \text{Post} + \gamma_1 X_{i,t} + \gamma_2 X_{i,t} \times \text{Post} + \delta_1 R_{ij,t} + \delta_2 R_{ij,t} \times \text{Post} + \mu_{j,t} + \eta_i + \epsilon_{ij,t} \quad (1)$$

where $\Delta b_{ij,t+1}$ is the difference in log credit granted by bank i to firm j between period t+1 and period t; $X_{i,t}$ are bank controls, $R_{ij,t}$ contains variables built using Credit Register (CR) information measuring the strength of firm-bank relation, such as share of *credit* and *bind* (see Section 3.2 for further details, or Table 1 for descriptions). Finally, $\mu_{i,t}$ and η_i are firm-time fixed effects and bank fixed effects. The variables $FVGovBonds_{i,pre}$ and $GovBonds_{i,pre}$ are absorbed by bank fixed effects because, as explained in the previous section, they are time-invariant in order to avoid the mechanical reduction of the ratio of government bonds held at fair value with respect to the ratio of government bonds in the post-shock period. The two variables of interest are $GovBonds_{i,pre} \times Post$ and $FVGovBonds_{i,pre} \times Post$: in particular, the coefficients α and β estimate respectively the contribution of the liquidity channel and the bank balance-sheet channel to the log-change in loans granted. To see why this is the case, consider first the ratio of government bonds held at amortised cost over total assets, $AmGovBonds_{i,pre}$; since changes in sovereign yields do not impact on the value of the bonds included in this portfolio, no decrease in the capital item is recorded and thus the bank balance sheet channel is not activated; on the contrary, the liquidity channel is triggered because when raising funds against collateral, the latter reflects market values despite the accounting method at which it is recorded in the bank's balance sheet. Therefore, $AmGovBonds_{i,pre}$ approximates the liquidity channel to which a bank is exposed to. Now consider eq.(1) and rearrange it as:

$$\begin{split} \Delta b_{ij,t+1} &= \alpha \text{GovBonds}_{i,pre} \times Post + \beta \text{FVGovBonds}_{i,pre} \times Post + \dots = \\ &= \alpha (\text{FVGovBonds}_{i,pre} + \text{AmGovBonds}_{i,pre}) \times Post + \beta \text{FVGovBonds}_{i,pre} \times \text{Post} + \dots = \\ &= \alpha \text{AmGovBonds}_{i,pre} \times \text{Post} + (\alpha + \beta) \text{FVGovBonds}_{i,pre} \times \text{Post} + \dots \end{split}$$

The coefficient on $GovBonds_{i,pre}$, α , is equal to the one that would be estimated if $AmGovBonds_{i,pre}$ were included in the regression and, as explained, $AmGovBonds_{i,pre}$ measures the liquidity channel. In the last equation, instead, the coefficient on $FVGovBonds_{i,pre}$ would estimate both the direct channels and it is equal to $(\alpha + \beta)$: given that α measures the effect of the liquidity channel, then it easily follows that β estimates the impact of the bank balance sheet channel.¹²

5 Results

Table 3 reports the estimates of the parameters of equation (1). As reference, Column 1 shows the results obtained with a specification without including the sovereign exposure held at fair value. Variables are lagged and the effect of sovereign tensions on the quarterly credit growth begins in Q3. The coefficient on $GovBonds_{i,pre} \times Post$ is found negative and significant, indicating that higher shares of public bond portfolios held before sovereign tensions are associated with larger reduction in bank credit supplied to firms in the post-shock period. Starting from Column 2 also sovereign exposure held at fair value is included. Column 2 reports the results without including bank fixed effects, so that the incidence of government bonds held in the pre-shock period can also be estimated: both $(GovBonds_{i,pre} \text{ and } FVGovBonds_{i,pre})$ are associated with higher growth of credit to firms; however, during the post-shock period, $GovBonds_{i,pre} \times Post$ is no longer significant and $FVGovBonds_{i,pre} \times Post$ becomes negative and significant; this suggests that the balance-sheet channel played a role for Italian banks during the second half of 2018, while the liquidity channel did not affect credit supply. Identification concerns might arise for estimates reported in Column 2, as the relative allocation of government bonds across different portfolios and banks' lending policies might be influenced by unobservable banking features, i.e. bank business model. To tackle this issue, Column 3 reports the results for the complete specification of equation (1), i.e. including bank fixed effects. In this specification $GovBonds_{i,pre} \times Post$ is no longer significant while $FVGovBonds_{i,pre} \times Post$ is negative and highly significant. This confirms that, even controlling for time-unvarying banks' characteristics, during the second half of 2018 (i.e. the post-shock period) the balance-sheet channel played a role and determined a reduction in credit supplied by Italian banks. The economic magnitude of the balance-sheet channel can be appraised by comparing credit supply for two banks differing in the share of government bonds held at fair value (before the shock) by a one-standard deviation (which corresponds to 4.56 pp). I find that a higher exposure to sovereign bonds at fair value determined a reduction in the growth rate of credit supply of 0.8 percentage points. This value is smaller if compared to findings of previous works: for example, De Marco (2018) finds that a one standard deviation increase in the market-to-market sovereign exposure is associated with a drop in the growth of credit of almost 6 percentage points; when recasting the estimates of Bottero et al. (2020) in percentage points, a one standard deviation increase in the sovereign to assets implies a reduction of around 4 percentage points.

The difference in the magnitude partly reflects the distinct nature of the event under study as De Marco (2018) and Bottero et al. (2020) focus on a period that radically changed the perception of government debt riskiness; however, such big gap suggests that during 2018

¹²The identification of the two channels may be weakened during prolonged and very acute phases of sovereign crises, when the market tends to value banks' public portfolios at fair value.

the effects of the *direct channels* was smaller than during 2011.

The coefficient on overall exposure to sovereign bonds - which measures the liquidity channel - is instead not statistically significant, suggesting that this channel did not activate at the aggregate level in the episode considered. It is still possible that the channel is activated for banks that have a structurally higher recourse to this source of funding. To take this into account, I add to equation (1) interactions with the dummy *high interbk_i*, which takes value 1 if, in the pre-shock period, the bank is above the median of the banklevel distribution of the variable *Interbank funding ratio* (corresponding to 15.4 per cent). Results, reported in Column 4, show that the liquidity channel did not activate also for this subset of banks, as $GovBonds_{i,pre} \times high interbk_t \times Post$ is not statistically significant (as $FVGovBonds_{i,pre} \times high interbk_i \times Post$). The remaining coefficients are unaffected and, in particular, the one on (the non-interacted) $FVGovBonds_{i,pre} \times Post$ remains negative and significant.

Results are robust to different way of clustering the standard errors: instead of double cluster at the bank and firm level, I cluster at the bank level (unreported regressions).

6 Extensions

In this section I extend the baseline model in two directions. First, I include controls for the other channels through which sovereign tensions transmit to the banking system. As mentioned, these are indirect channels and tend to activate - during periods of turbulence on the sovereign debt market - irrespective of a given bank's direct exposure to public debt. This analysis has the twofold objective of being a robustness check for the baseline results and of providing an assessment of the relative importance of the direct channels as compared to the other transmission channels of sovereign tensions.

Second, I dig deeper into the result on the (non-activation of the) liquidity channel and, in particular, check whether this result may be explained by the very broad availability of funds raised via the Eurosystem's refinancing operations. This hypothesis would reconcile the result of the irrelevance of the liquidity channel in 2018 with those of the studies on the sovereign debt crisis, according to which the dry-up of funding liquidity for banks was arguably the main factor behind the tightening of loan supply conditions since the end of 2011.

6.1 Taking into account the indirect channels of sovereign tensions

The baseline model presented in Section 4 focuses on the transmission channels - from sovereign tensions to bank balance sheets and thus bank lending - activated by the direct holdings of government debt by credit institutions (*direct channels* of sovereign tensions). As mentioned, however, a rise in sovereign yields might impact banks and thus lending policies through several other indirect channels (for a detailed review of the transmission channels of sovereign tensions, see BIS 2011). First, a prominent indirect channel is the generalized increase in the cost of funding for banks that is typically associated with a rise in sovereign yields (which are the basis for the pricing of bank bonds¹³) and which is transmitted to the private sector via a worsening of lending conditions (*cost of funding channel*; Angelini et al. 2014, Bofondi et al. 2017). One way to assess the impact of sovereign tensions via this channel is to use a proxy of banks' refinancing needs in the post-shock period, as credit institutions will face an increased cost to roll-over debt expiring during the post-shock period. To this end, I include in the regression the amount of bonds issued that are due to expire in the second half of 2018 over total assets, that is *Maturing issued bonds ratio*.

Second, sovereign tensions may be transmitted to banks also via a deterioration of the implicit or explicit guarantees that governments provide to credit institutions, should they end up in financial distress (government guarantee channel). Banks with larger amounts of deposits held by households and firms tend to be more affected by this channel, as these institutions are *ex ante* regarded as more likely to be bailed-out by authorities in order to protect deposit holders. Following Mäkinen et al. (2020), I account for this transmission channel by including in the regressions the variable *Deposit retail ratio*, that is the share of deposits held by households over the GDP, which measures a given bank's market share of deposits and is positively correlated with the expected support of government in case of bank distress.

Third, following a rise in government yields the amount of credit granted might reduce as a consequence of banks' decision to modify the composition of their assets and increase the amount of public bonds (crowding-out effect of lending). This phenomenon is documented by several studies (see Battistini et al. 2014) and, differently from the other channels described so far, is not associated with a deterioration of banks' balance sheet conditions. Some studies explain it as the result of pressure by the governments on credit institutions to buy domestic debt during periods of sovereign stress (*moral suasion* hypothesis, see Becker and Ivashina 2018 and Ongena et al. 2019). For others this phenomenon is driven by profitability and risk-shifting motives (see Acharya and Steffen 2015). In order to control for this effect, I augment equation (1) with the variable *Government bonds purchases*, which measures the growth of sovereign bonds net purchases at the bank level. Notice that - in order to capture potential crowding out effects - this variable is included with the same timing of the dependent variable (and not with a 1-period lag).

Two additional indirect channels, which are often mentioned as important in the sovereignbank nexus, are not included in this analysis because they most likely did not play a relevant role in the episode considered. First, I do not consider the *sovereign downgrade channel*, which arises as sovereign ratings typically represent a ceiling for those assigned to private borrowers: a sovereign downgrade is often followed by downgrades of credit institutions, which increases the cost and reduces availability of funding on financial markets. In the period object of this study this channel is unlikely to have played any role, as only Moody's

 $^{^{13}{\}rm For}$ more details, see the ECB website The compass of monetary policy: favourable financing conditions, Chart 4.

- among the main credit rating agencies - downgraded the Italian sovereign rating in the course of 2018 by one single notch, to Baa3, without affecting its investment-grade class. Moreover, the downgrading occurred at the end of October 2018, which is almost at the very end of the post-shock period included in the regression. Second, sovereign tensions can be transmitted via international spillovers, for example through cross-border interbank exposure or through direct claims vis-à-vis the non-financial sector of countries in distress. As the episode of sovereign tensions in 2018 was entirely driven by the uncertainty related to the Italian political situation and other countries were not affected, international spillovers hardly played any role.

Results for the extended regressions are presented in Table 4. The first 3 columns include the three variables capturing the indirect channels one at the time: *Maturing issued bonds ratio*, *Deposit retail ratio* and *Government bond purchases*; the last column includes all of them. For the direct channels, results are always consistent with those of the baseline regression: in each specification the coefficient on the balance-sheet channel is negative and significant and implies a lower growth rate for highly-exposed banks by about 0.8 percentage points (1.0 for the specification in column 2). Similarly, the coefficient on the liquidity channel is not statistically different from zero.

The table shows that sovereign tensions affected credit supply also through all the indirect channels, as the coefficients on all the three proxies of these indirect channels are negative and significant when they are included one at the time; in Column 4, instead, *Deposit retail* \times *Post* is no longer significant, signalling that in 2018 the government guarantee channel did not play a role. In terms of magnitude, the impact of the cost of funding is sizeable and comparable to that of the balance-sheet channel: the difference in credit supply growth for two banks differing by a one-standard deviation of the *Maturing issues bonds ratio* \times *Post*, is 1.1 percentage points. This result is overall consistent with previous research conducted on the sovereign debt crisis: as for instance, Bofondi et al. (2017) also find that, in the semester following the abrupt rise in Italian government yields at the beginning of July 2011, domestic banks displaying a growth of credit smaller than 3 percentage points with respect to that of foreign branches. Finally, the variable *Government bond purchases* is also found negative and significant, but the economic magnitude is negligible.

6.2 Private markets and Eurosystem operations

The baseline results indicate that the deterioration in banks' liquidity position following the sovereign shock did not contribute to reduce lending supply. This finding is somewhat at odds with those obtained in the literature on the sovereign debt crisis, which identified in the dry-up of liquidity one of the main channels of transmission of sovereign tensions to banks (see, for instance, Angelini et al. 2014). The exercise carried out in the previous section partly reconciles this apparently divergent result as it documents that the indirect cost of funding channel plays a role in explaining the reduction in credit supply - moreover, its economic magnitude is comparable to that associated with the *balance-sheet channel*. In this section I focus instead on the direct channels and investigate whether a potential explanation behind the difference in the results compared to the studies analysing the sovereign debt crisis could be related to the overall stronger liquidity position of the banking system in 2018 following several years of expansionary monetary policy by the ECB. In particular, since the 3-year long term refinancing operations (LTRO) launched in December 2011, the recourse to Eurosystem refinancing markedly increased - and, consequently, there was a decline in the relative share of collateralized funds raised on the private markets. In April 2018, just before the abrupt increase in the Italian government yields, the share of Eurosystem funds over total collateralized funds was 23.4 per cent, compared to less than 6 per cent in June 2011, just before the sovereign debt crisis hit the Italian banking system (Fig.3). I conjecture that the increase in the share of funds raised via Eurosystem operations played a role in neutralizing the liquidity channel in 2018.

Banks that normally raise funds more extensively on the wholesale markets rather than via Eurosystem operations might have experienced a stronger deterioration in the ability of raising funds: indeed, private counterparts can trigger margin calls or apply larger haircuts following an increase in sovereign risk; on the contrary, haircuts applied on refinancing operations by Eurosystem national central banks depend only on the residual maturity and the rating of the government bond posted as collateral.¹⁴ Therefore, during sovereign tensions, the larger the recourse to the private markets (with respect to the Eurosystem operations) the stronger the liquidity shock hitting credit institutions.

To test this hypothesis, I proceed in the following way: first, for each bank I compute the share of total funds raised on private collateralized markets (with either euro area private banks or central counterparties) over the total collateralized funding (which comprises also the liquidity raised via Eurosystem operations):

$$share_mkt = \left(\frac{Privatewholesalefunding_{i,pre}}{Totalwholesalefunding_{i,pre}}\right)$$

I then interact share $_mkt$ with GovBonds in eq. (1) and estimate the following regression:

$$\Delta b_{ij,t+1} = \theta(GovBonds_{i,pre} \times share_mkt \times Post) + \alpha GovBonds_{i,pre} \times Post + \beta FVGovBonds_{i,pre} \times Post +$$
(2)

The hypothesis being tested here is that larger recourse to private collateralized markets with respect to Eurosystem operations are associated with an activation of the *liquidity* channel, ceteris paribus; a larger share_mkt is supposed to have a negative impact on credit supply and as a consequence, the coefficient on the triple interaction term in the brackets is expected to be negative ($\theta < 0$).

¹⁴For details, see on the ECB website Financial Risk Management of Eurosystem Monetary Policy Operations, July 2015.

Results for equation (3) are shown in Table 5. The specification of Column 1 does not include FVGovBonds, similarly to Table 3. The coefficient on the triple interaction term is negative and significant, suggesting that the liquidity channel is activated as the recourse to the private collateralized markets increases. The coefficient on *GovBonds* is negative, even though is not significant. Column 2 includes the share of government bonds held at fair value among the independent variables: consistently with the findings of Section 5, FVGovBonds is found negative and significant. Remarkably, also the coefficient on the triple interaction remains negative and significant, suggesting that the *liquidity channel* plays a role for the more exposed banks on the private markets even when controlling for the bank balance-sheet channel. Turning to the economic magnitude, for the same borrower the growth of credit granted by a hypothetical bank that raises funds on the private markets only (share = 1) is lower by 0.3 percentage points than the growth of credit granted by a bank that raises funds only via Eurosystem operations (share mkt = 0). This result partly reconciles the findings of Table 3 with previous studies on the sovereign debt crisis, documenting that the *liquidity channel* plays a role even outside periods of severe financial strains as in 2018.¹⁵ Finally, for completeness Column 3 reports the estimates of the regression when adding also the proxies for the indirect channels described in the previous subsection (Maturing issued bonds ratio \times Post, Deposit retail ratio \times Post and Government bonds purchases \times Post): the evidence is in line with Table 3, and indicates an activation of both the balance-sheet and the cost of funding channels.

7 Concluding Remarks

In this paper I provide a quantitative estimation of the relative importance of the *balance-sheet channel* and the *liquidity channel* focusing on the development of credit supply for Italian banks in the second half of 2018. The period under study follows the exogenous increase in the Italian sovereign yields observed in May and connected to the high uncertainty over the formation of the new government. I find that banks' direct holdings of sovereign bonds had a role in transmitting the sovereign shock to credit supply: in particular, the reduction in lending supply was the consequence of the adverse shock to banks' capital position, while the liquidity channel did not contribute. A further extension suggests that the non-activation of the liquidity channel may be related to the very large availability of funds granted via Eurosystem operations in the horizon considered. Finally, the paper confirms that the generalized increase in banks' funding cost associated with the rise in sovereign yields also played a role in the reduction in credit supply.

¹⁵As the regression always includes *interbank_ratio* and *size* among controls, *share_mkt* captures different compositions of the wholesale funding. However, in order to take into account the absolute exposition of the bank to the private markets, as a robustness check I run again eq. (3) by substituting *share_mkt* first with (i) *mkt_ratio* (the funds raised on the private markets over total assets) and, then, with (ii) ECB_ratio (the funds raised via Eurosystem operations over total assets). Table A1 in the Appendix reports the results. As it shows, only the triple interaction with *mkt_ratio* is negative and significant, while that with ECB_ratio is not: this is in line with Table 5, and confirms that larger exposures to the private markets are associated with the activation of the liquidity channel.

All in all, results are in line with previous research that finds a negative association between total direct holdings of sovereign bonds and credit supply during episodes of a rise in government yields (Bottero et al. 2020). At the same time the evidence presented in this paper suggests that also the *cost of funding channel* played a comparable role to that of the holdings of sovereign bonds; this in line with the literature documenting that the association between the total direct holdings of public bonds and credit supply may vanish during very acute phases of a crisis, as in this case the impact on banks is mainly driven by a country-specific effect and is largely independent of the composition of their portfolios (Bofondi et al. 2017).

As an important *caveat*, this analysis does not take into account that the direct holdings of government bonds by credit institutions also have important beneficial effects for financial and, in turn, macroeconomic stability. This is for example related to banks' tendency to act as contrarian investors in the sovereign bond markets during episodes of financial tensions (Lanotte et al. 2016).

Assessing whether sovereign tensions are transmitted prominently via direct or indirect channels is a crucial information for policymakers: future research could therefore aim at comparing the overall effects of direct channels with those of indirect channels of sovereign tensions, possibly focusing not only on different episodes of rises of government yields, but also on the effects of different accountability rules in place when the sovereign tensions occur.

References

- Acharya, V. and Steffen, S. (2015). The "greatest" carry trade ever? understanding eurozone bank risks. Journal of Financial Economics, 115(2):215–236.
- Acharya, V. V., Eisert, T., Eufinger, C., and Hirsch, C. (2019). Whatever It Takes: The Real Effects of Unconventional Monetary Policy. *Review of Financial Studies*, 32(9):3366–3411.
- Adelino, M. and Ferreira, M. A. (2016). Bank Ratings and Lending Supply: Evidence from Sovereign Downgrades. *Review of Financial Studies*, 29(7):1709–1746.
- Albertazzi, U., Ropele, T., Sene, G., and Signoretti, F. M. (2014). The impact of the sovereign debt crisis on the activity of Italian banks. *Journal of Banking & Finance*, 46:387–402.
- Altavilla, C., Pagano, M., and Simonelli, S. (2015). Bank Exposures and Sovereign Stress Transmission. CSEF Working Papers 410, Centre for Studies in Economics and Finance (CSEF), University of Naples, Italy.
- Angelini, P., Grande, G., and Panetta, F. (2014). The negative feedback loop between banks and sovereigns. Questioni di Economia e Finanza (Occasional Papers) 213, Bank of Italy, Economic Research and International Relations Area.
- Arteta, C. and Hale, G. (2008). Sovereign debt crises and credit to the private sector. Journal of International Economics, 74(1):53–69.
- Baskaya, Y. S., di Giovanni, J., Kalemli-Özcan, S., Peydro, J.-L., and Ulu, M. F. (2017). Capital flows and the international credit channel. *Journal of International Economics*, 108:S15–S22.
- Battistini, N., Pagano, M., and Simonelli, S. (2014). Systemic risk, sovereign yields and bank exposures in the euro crisis. *Economic Policy*, 29(78):203–251.
- Becker, B. and Ivashina, V. (2018). Financial Repression in the European Sovereign Debt Crisis. *Review of Finance*, 22(1):83–115.
- Bernanke, B. and Lown, C. S. (1991). The credit crunch. Brookings Papers on Economic Activity, 22(2):205–248.
- Bernanke, B. S. and Gertler, M. (1995). Inside the Black Box: The Credit Channel of Monetary Policy Transmission. Journal of Economic Perspectives, 9(4):27–48.
- BIS (2011). The impact of sovereign credit risk on bank funding conditions. CGFS Papers, Bank for International Settlements. 43.
- Bofondi, M., Carpinelli, L., and Sette, E. (2017). Credit Supply During a Sovereign Debt Crisis. Journal of the European Economic Association, 16(3):696–729.
- Bottero, M., Lenzu, S., and Mezzanotti, F. (2020). Sovereign debt exposure and the bank lending channel: Impact on credit supply and the real economy. *Journal of International Economics*, 126. Forthcoming.

- Cetorelli, N. and Goldberg, L. (2011). Global banks and international shock transmission: Evidence from the crisis. *IMF Economic Review*, 59(1):41–76.
- Correa, R., Lee, K.-H., Sapriza, H., and Suarez, G. A. (2014). Sovereign Credit Risk, Banks' Government Support, and Bank Stock Returns around the World. *Journal of Money, Credit and Banking*, 46(s1):93–121.
- De Marco, F. (2018). Bank lending and the european sovereign debt crisis. *Journal of Financial and Quantitative Analysis*, 54:1–28.
- Gennaioli, N., Martin, A., and Rossi, S. (2018). Banks, government bonds, and default: What do the data say? *Journal of Monetary Economics*, 98(C):98–113.
- Jimenèz, G., Ongena, S., Peydrò, J.-L., and Saurina, J. (2012). Credit supply and monetary policy: Identifying the bank balance-sheet channel with loan applications. *American Economic Review*, 102(5):2301–26.
- Jiménez, G., Ongena, S., Peydró, J.-L., and Saurina, J. (2014). Hazardous times for monetary policy: What do twenty-three million bank loans say about the effects of monetary policy on credit risk-taking? *Econometrica*, 82(2):463–505.
- Kashyap, A. K. and Stein, J. C. (1995). The impact of monetary policy on bank balance sheets. Carnegie-Rochester Conference Series on Public Policy, 42:151–195.
- Khwaja, A. I. and Mian, A. (2008). Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market. American Economic Review, 98(4):1413–1442.
- Lanotte, M., Manzelli, G., Rinaldi, A. M., Taboga, M., and Tommasino, P. (2016). Easier said than done? Reforming the prudential treatment of banks' sovereign exposures. Questioni di Economia e Finanza (Occasional Papers) 326, Bank of Italy, Economic Research and International Relations Area.
- Mäkinen, T., Sarno, L., and Zinna, G. (2020). Risky bank guarantees. Journal of Financial Economics, 136(2):490–522.
- Olivier, D. J., Hans, D., Klaas, M., Ongena, S., and Glenn, S. (2020). Some Borrowers Are More Equal than Others: Bank Funding Shocks and Credit Reallocation. *Review of Finance*, 24(1):1–43.
- Ongena, S., Popov, A., and Van Horen, N. (2019). The invisible hand of the government: Moral suasion during the european sovereign debt crisis. *American Economic Journal: Macroeconomics*, 11(4):346–79.
- Peek, J. and Rosengren, E. (1997). The international transmission of financial shocks: The case of japan. American Economic Review, 87(4):495–505.
- Peek, J. and Rosengren, E. S. (2000). Collateral damage: Effects of the japanese bank crisis on real activity in the united states. *American Economic Review*, 90(1):30–45.
- Popov, A. and Van Horen, N. (2014). Exporting Sovereign Stress: Evidence from Syndicated Bank Lending during the Euro Area Sovereign Debt Crisis. *Review of Finance*, 19(5):1825–1866.
- Puri, M., Rocholl, J., and Steffen, S. (2011). Global retail lending in the aftermath of the us financial crisis: Distinguishing between supply and demand effects. *Journal of Financial Economics*, 100(3):556–578.

- Schivardi, F., Sette, E., and Tabellini, G. (2017). Credit misallocation during the european financial crisis. Temi di discussione (Economic working papers) 1139, Bank of Italy, Economic Research and International Relations Area.
- Schnabl, P. (2012). The International Transmission of Bank Liquidity Shocks: Evidence from an Emerging Market. *Journal of Finance*, 67(3):897–932.
- Stein, J. C. and Kashyap, A. K. (2000). What Do a Million Observations on Banks Say about the Transmission of Monetary Policy? *American Economic Review*, 90(3):407– 428.

Tables and figures



Figure 1: Government bonds, by bank type

Source: Based on Supervisory Reports statistics.

The figure shows the incidence, over total assets, of Italian public bonds (blue histogram) of which held at fair value (purple histogram) as reported in Supervisory Reports statistics for March 2018.



Figure 2: Yields on Italian government bonds

Source: Refinitiv.



Figure 3: Funding on collateralized markets by Italian banks

The figure splits the total of collateralized funds into (i) funds raised toward ECB operations (blu area) and (ii) funds raised on private markets (orange area), as observed in two points in time just preceding the sharp increase in Italian government yields related to the sovereign debt crisis occurred (June 2011) and the one occurred in 2018 related to the political uncertainty in Italy at the time (April 2018). Both histograms sum to 100 per cent. Data are from Supervisory Reports statistics.



Figure 4: Italian banks and sovereign CDS

Each point corresponds to the daily observation of the sovereign 5-year CDS for Italy (x-axis) and 5-year CDS for the Italian banking sector (y-axis). The 5-year CDS for the banking sector is computed as the unweighted mean of the CDS for the Italian credit institutions with a CDS. Daily data from Refinitiv.

Variable	Description
Growth of credit	Log diff. of credit granted from bank to firm c horizon of 3 months
Capital ratio	Tier 1 ratio (for the post-shock period, amen for the changes due to sovereign tensions)
Liquidity ratio	Cash and foreign sovereign bonds over total as Funds raised on wholesale markets, with eit
Interbank funding ratio	private counterparties or with European Cen Bank, over total assets
ROA	Profit and losses over total assets
Size	Log of total assets
Maturing bonds ratio	Issued bonds with residual maturity up to Dece ber 2018 over total assets
Government Bonds ratio	Italian sovereign bonds over total assets
Government Bonds at fair value ratio	Italian sovereign bonds held in the fair value perfolios over total assets
Retail deposits to GDP	Deposits held by households over to nominal nual GDP
Government bonds purchases	Growth of purchases of government bonds over period
Share of credit	Credit obtained by firm j from bank i over to bank loans obtained by firm i
Bind	The difference between credit granted and cred drawn over total credit granted

Table 1 -	Description	of variables
-----------	-------------	--------------

Table 2 - Des	scriptive	statistics					
Variable	Mean	Median	25pct	75pct	Std.Dev.		
Panel A : Pre-shock							
Growth of credit	-0.6	0	-3.89	0	21.91		
Capital ratio	16.14	15.12	14.08	17.8	2.66		
Liquidity ratio	1.62	1.84	0.51	2.42	1.17		
Interbank funding ratio	13.65	12.13	10.35	15.97	6.73		
ROA	0.28	0.17	0.12	0.49	0.44		
Size (log. assets)	10.97	11.32	9.673	12.29	2.212		
Maturing issued bonds ratio	1.6	1.5	0.65	2.65	1.19		
Government bonds ratio	10.34	7.78	5.47	14.33	7.52		
Government bonds at fair value ratio	5.97	4.21	2.4	7.28	4.56		
Retail deposits to GDP ratio	2.66	1.44	0.21	2.92	3.14		
Government bonds purchases	6.98	8.11	-9.7	21.91	68.09		
Share of credit	32	26	14	46	22		
Bind	41	36	8	72	44		
Variable	Mean	Median	25pct	75pct	Std.Dev.		

Table 2 - Descriptive statistics

Panel B: Post-shock

Growth of credit	-0.91	-0.07	-4.31	0	22.23
Capital ratio	16.54	15.36	13.41	18.4	3.92
Liquidity ratio	1.89	1.46	0.64	2.9	1.27
Interbank funding ratio	15.3	15.74	10.31	18.31	7.43
ROA	0.27	0.36	0.21	0.48	0.43
Size (log. assets)	11.22	11.83	9.871	13.11	2.173
Maturing issued bonds ratio	1.19	1.04	0.44	1.65	0.91
Government bonds ratio	11.28	9.03	5.67	14.33	8.06
Government bonds at fair value ratio	5.61	3.99	2.16	8.26	4.08
Retail deposits to GDP ratio	3.26	1.86	0.32	7.14	3.31
Government bonds purchases	0.03	-0.5	-6.9	2	21.35
Share of credit	32	27	14	46	22
Bind	42	37	9	74	54

The table shows descriptive statistics of the variables used in the regressions, relative to the estimating sample and thus taken over the distribution of firm-bank-quarter observations. Data for Growth of credit, Share of credit and Bind are from the Italian Credit Register. Data on bank characteristics are from the Supervisory Reports. The sample period includes bank-firm relationships observed in March, June, September and December 2018. Growth of credit is the difference in the quarterly log credit granted. Capital ratio is the ratio of regulatory capital to risk weighted assets; liquidity ratio is the ratio of cash plus non-domestic government securities to total assets; interbank funding ratio is the ratio of wholesale deposits to total assets; ROA is the ratio of bank profit t to total assets; bank size is the log of total assets; maturing issued bonds is the ratio of bonds issued and with residual maturity up to December 2018 to total assets; government bonds ratio is the ratio between the amount of Italian public bonds over total assets; government bonds held at fair value ratio is the ratio between the amount of Italian public bonds allocated under either the fair value through profit and loss portfolio or the fair value through other comprehensive income portfolio over total assets; retail deposits to GDP ratio is the ratio between deposits held by households over the annual nominal GDP; government bonds purchases is the quarterly growth of purchases in Italian public bonds; share of credit is the credit obtained by a firm from the bank over total bank loans held by the firm; Bind is the ratio between the difference of credit granted and credit drawn over the total credit granted.

VARIABLES	(1)	(2)	(3)	(4)
GovBonds		0.14***		
FVGovBonds		(4.56) 0.13^{***}		
GovBonds x Post	-0.12**	(2.89) -0.03	-0.06	0.02
FVGovBonds x Post	(-2.01)	(-0.54) -0.26***		
GovBonds x HighInterbk x Post		(-4.06)	(-2.49)	(-1.90) -0.10
FVGovBonds x HighInterbk x Post				(-0.97) -0.16 (-1.33)
Observations	1,051,388	1,051,389	1,051,388	1,051,388
R-squared	0.389	0.387	0.389	0.389
Bank controls	yes	yes	yes	yes
Bank fixed effects	yes	no	yes	yes
Firm [*] quarter fixed effects	yes	yes	yes	yes

Table 3 - The channels of sovereign tensions on credit supply: disentangling the direct channels

The table shows regressions of the change in the log of credit granted over a 3-month horizon on the interaction between (i) the share of Italian government bonds over total assets (GovBonds) and the dummy Post, (ii) the share of Italian government bonds held at fair value (FVGovBonds) and the dummy Post. Bank controls and variables at firm-bank level and all their interactions with the dummy Post are added in every regression. To avoid measurement errors arising by the different impact of a rise in sovereign yields on GovBonds and FVGovBonds, these two variables are measured at March 2018 and are therefore time-invariant. The dependent variable is computed as the change in the log of credit granted by the bank to the firm in period t and period t+1. Firm and bank level controls are measured at the end of period t. All independent variables are lagged with respect to the dependent variable. The dummy Post is equal to one if the period t is Jun-2018 or later. The regression includes one pre-shock period and two post-shock periods. The last column also includes the interaction of a dummy equal to one if the bank is above the median in terms of total interbank funding over total assets. Standard errors are double clustered at the bank and firm level. Robust t-statistics in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

VARIABLES	(1)	(2)	(3)	(4)
GovBonds x Post	-0.07	-0.06	-0.05	-0.09
	(-1.44)	(-1.30)	(-0.90)	(-1.58)
FVGovBonds x Post	-0.16*	-0.21***	-0.19**	-0.15*
	(-1.87)	(-3.44)	(-2.44)	(-1.90)
Maturing issued bonds ratio x Post	-0.57**	· · · ·	· · ·	-0.83**
	(-1.99)			(-2.52)
Deposit retail x Post	()	-0.46*		-0.24
		(-1.84)		(-1.05)
Government bonds purchases x Post		× /	-0.02**	-0.02**
			(-2.27)	(-2.26)
Observations	1,051,388	1,051,388	1,046,642	1,046,642
R-squared	0.389	0.389	0.389	0.389
Bank controls	yes	yes	yes	yes
Bank fixed effects	yes	yes	yes	yes
Firm*quarter fixed effects	yes	yes	yes	yes

Table 4 - The channels of sovereign tensions on credit supply: controlling for other indirect channels

The table shows regressions of the change in the log of credit granted over a 3-month horizon on the interactions between (i) the share of Italian government bonds over total assets (GovBonds) and the dummy Post, (ii) the share of Italian government bonds held at fair value (FVGovBonds) and the dummy Post. Bank controls and variables at firm-bank level and all their interactions with the dummy Post are added in every regression. To avoid measurement errors arising by the different impact of a rise in sovereign yields on GovBonds and FVGovBonds, these two variables are measured at March 2018 and are therefore time-invariant. The dependent variable is computed as the change in the log of credit granted by the bank to the firm in period t and period t+1. Firm and bank level controls are measured at the end of period t. The dummy Post is equal to one if the period t is Jun-2018 or later. The regression includes one pre-shock period and two post-shock periods. Maturing issued bonds ratio is the ratio, over total assets, of bonds issued with maturity up to December 2018 and proxies the cost of funding channel. Deposit retail ratio is computed as bank deposits held by households over GDP and controls for the government guarantee channel. Government bonds purchases is the growth in the domestic public bonds purchases, controlling for the crowding-out effect of lending, is the only variable not lagged with respect to the dependent variable. Standard errors are double clustered at the bank and firm level. Robust t-statistics in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

VARIABLES	(1)	(2)	(3)
GovBonds x Share_mkt x Post	-0.26*	-0.28**	-0.36***
Shana mht y Daat	(-1.88) 0.03	(-2.13) 0.03	(-2.75) 0.06^{**}
Share_mkt x Post	(1.11)		(2.08)
GovBonds x Post	-0.03	0.05	0.03
FVGovBonds x Post	(-0.36)	(0.75) - 0.21^{***}	
Deposit retail x Post		(-3.11)	(-2.36) -0.30
-			(-1.14) -0.84***
Maturing issued bonds ratio x Post			(-2.75)
Government bonds purchases x Post			-0.01**
			(-2.23)
Observations	$1,\!050,\!950$	$1,\!050,\!950$	1,046,348
R-squared	0.389	0.389	0.390
Bank controls	yes	yes	yes
Bank fixed effects	yes	yes	yes
Firm*quarter fixed effects	yes	yes	yes

Table 5 - The role of recourse to Eurosystem funds on the activation of the liquidity channel

The table shows regressions of the change in the log of credit granted over a 3-month horizon on the interactions between (i) the share of Italian government bonds over total assets and the dummy Post, (ii) the share of Italian government bonds over total assets scaled on the relative recourse to funds raised on private collateralized markets over total wholesale funding (share mkt) and the dummy Post and (iii) the share of Italian government bonds held at fair value (FVGovBonds) and the dummy Post. Bank controls and variables at firm-bank level and all their interactions with the dummy Post are added in every regression. Share mkt is computed as the share of funds raised on the private collateralized markets over the total funds raised on the collateralized markets. To avoid measurement errors arising by the different impact of a rise in sovereign yields on GovBonds and FVGovBonds, these two variables are measured at March 2018 and are therefore time-invariant. The dependent variable is computed as the change in the log of credit granted by the bank to the firm in period t and period t+1. Firm and bank level controls are measured at the end of period t. The dummy Post is equal to one if the period t is Jun-2018 or later. The regression includes one pre-shock period and two post-shock periods. Maturing issued bonds ratio is the ratio, over total assets, of bonds issued with maturity up to December 2018 and proxies the cost of funding channel. Deposit retail ratio is computed as bank deposit held by households over GDP and controls for the government guarantee channel. Government bonds purchases is the growth in the domestic public bonds purchases, controlling for the crowding-out effect of lending, is the only variable not lagged with respect to the dependent variable. Standard errors are double clustered at the bank and firm level. Robust t-statistics in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

Appendix

	(1)	(2)	(3)	(4)
VARIABLES				
GovBonds x Post	-0.05	0.05	-0.05	-0.01
	(-0.85)	(0.85)	(-0.78)	(-0.10)
GovBonds x mkt_ratio x Post	-1.35*	-1.71**		
	(-1.88)	(-2.22)		
mkt ratio x Post	0.10	0.16		
—	(1.08)	(1.51)		
FVGovBonds x Post	× ,	-0.25***		-0.21***
		(-4.92)		(-4.37)
GovBonds x ECB_ratio xPost		· · · ·	-1.35	-0.95
—			(-1.50)	(-1.16)
ECB ratio x Post			0.30*	0.24
_			(1.68)	(1.42)
Observations	1,050,993	1,050,993	1,050,993	1,050,993
R-squared	0.389	0.389	0.389	0.389
Bank controls	yes	yes	yes	yes
Bank fixed effects	yes	yes	yes	yes
Firm [*] quarter fixed effects	yes	yes	yes	yes

Table A1 - The role of recourse to Eurosystem funds on the activation of the liquidity channel: different definitions for share mkt

The table is a robustness check for Table 5. In Columns 1 and 2 the change in the log of credit granted over a 3-month horizon is regressed on the interactions between (i) the share of Italian government bonds over total assets and the dummy Post, (ii) the share of Italian government bonds over total assets, the relative recourse to funds raised on private collateralized markets over total assets (mkt ratio) and the dummy Post and (iii) the share of Italian government bonds held at fair value (FVGovBonds) and the dummy Post. In Columns 3 and 4 the change in the log of credit granted over a 3-month horizon is regressed on the interactions between (i) the share of Italian government bonds over total assets and the dummy Post, (i) the share of Italian government bonds over total assets, the relative recourse to funds raised via Eurosystem operations over total assets (ECB ratio) and the dummy Post and (iii) the share of Italian government bonds held at fair value (FVGovBonds) and the dummy Post. Bank controls and variables at firm-bank level and all their interactions with the dummy Post are added in every regression. To avoid measurement errors arising by the different impact of a rise in sovereign yields on GovBonds and FVGovBonds, these two variables are measured at March 2018 and are time-invariant. The dependent variable is computed as the change in the log of credit granted by the bank to the firm in period t and period t+1. Firm and bank level controls are measured at the end of period t. The dummy Post is equal to one if the period t is Jun-2018 or later. The regression includes one preshock period and two post-shock periods. Standard errors are double clustered at the bank and firm level. Robust t-statistics in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1375 New facts on consumer price rigidity in the Euro Area, by Erwan Gautier, Cristina Conflitti, Riemer P. Faber, Brian Fabo, Ludmila Fadejeva, Valentin Jouvanceau, Jan-Oliver Menz, Teresa Messner, Pavlos Petroulas, Pau Roldan-Blanco, Fabio Rumler, Sergio Santork, Elisabeth Wieland and Hélène Zimmer (July 2022).
- N. 1376 *Connecting to power: political connections, innovation, and firm dynamics,* by Ufuk Akcigit, Salomé Baslandze and Francesca Lotti (July 2022).
- N.1377 "Green" fiscal policy measures and non-standard monetary policy in the Euro Area, by Anna Bartocci, Alessandro Notarpietro and Massimiliano Pisani (July 2022).
- N.1378 *Firm liquidity and the transmission of monetary policy*, by Margherita Bottero and Stefano Schiaffi (July 2022).
- N.1379 *The impact of Covid-19 on the european short-term rental market*, by Elisa Guglielminetti, Michele Loberto and Alessandro Mistretta (July 2022).
- N.1380 An analysis of objective inflation expectations and inflation risk premia, by Sara Cecchetti, Adriana Grasso and Marcello Pericoli (July 2022).
- N. 1381 The role of central bank communication in inflation-targeting Eastern European emerging economies, by Valerio Astuti, Alessio Ciarlone and Alberto Coco (October 2022).
- N. 1382 The heterogeneous effects of bank mergers and acquisitions on credit to firms: evidence from Italian macro-regions, by Silvia Del Prete, Cristina Demma, Iconio Garrì, Marco Piazza and Giovanni Soggia (October 2022).
- N. 1383 Bank lending to small firms: metamorphosis of a financing model, by Paolo Finaldi Russo, Valentina Nigro and Sabrina Pastorelli (October 2022).
- N. 1384 MMacroeconomic effects of growth-enhancing measures in the euro area, by Alessandro Cantelmo, Alessandro Notarpietro and Massimiliano Pisani (October 2022).
- N.1385 *Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel*, by Claire Giordano (October 2022).
- N. 1386 Labor market spillovers of a large plant opening. Evidence from the oil industry, by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani (October 2022).
- N. 1387 Unburdening regulation: the impact of regulatory simplification on photovoltaic adoption in Italy, by Federica Daniele, Alessandra Pasquini, Stefano Clò and Enza Maltese (October 2022).
- N.1388 *Liberalizing the opening of new pharmacies and hospitalizations*, by Andrea Cintolesi and Andrea Riganti (November 2022).
- N.1389 *Mediation Analysis Synthetic Control*, by Giovanni Mellace and Alessandra Pasquini (November 2022).
- N.1390 The effects of partial employment protection reforms: evidence from Italy, by Diego Daruich, Sabrina Di Addario and Raffaele Saggio (November 2022).
- N. 1391 *The role of majority status in close election studies*, by Matteo Alpino and Marta Crispino (November 2022).
- N.1392 Issuing bonds during the Covid-19 pandemic: is there an ESG premium?, by Fabrizio Ferriani (November 2022).
- N. 1393 *Economic fundamentals and stock market valuation: a CAPE-based approach*, by Maria Ludovica Drudi and Federico Calogero Nucera (November 2022).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times,* R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities,* Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014).**
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy, Journal of International Economics, v. 126, article 103328, WP 1032 (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, *Venture capitalists at work: a Diff-in-Diff approach at late-stages of the screening process*, Journal of Business Venturing, v. 35, 3, **WP 1131 (September 2017).**
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis,* Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DE BLASIO G., I. DE ANGELIS and L. RIZZICA, *Lost in corruption. Evidence from EU funding to Southern Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 6, 3, pp. 355–377, WP 1180 (December 2016).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, WP 1299 (October 2020).
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, Article 103201, WP 1142 (October 2017).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- RIZZICA L. and M. TONELLO, Persuadable perceptions: the effect of media content on beliefs about

"TEMI" LATER PUBLISHED ELSEWHERE

corruption, Economic Policy, v. 35, 104, pp. 679-737, WP 1043 (November 2016).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, **WP 1072 (July 2016).**

2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019)**.
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno*, Papers in Regional Science, v. 100, 2, pp. 327-348, **WP 1253 (December 2019).**
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, **WP 1133 (September 2017).**
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, Intergenerational mobility in the very long run: Florence 1427-2011, Review of Economic Studies, v. 88, 4, pp. 1863–1891, WP 1060 (April 2016).BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (Oct 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, WP 1288 (July 2020).
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).

- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- DEL PRETE S. and M. L. STEFANI, *Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks*, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, **WP 1014 (June 2015).**
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area,* Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, WP 1101 (February 2017).
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, **WP 1277 (June 2020).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, **WP 1095 (January 2017).**
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., *Demographics and the natural real interest rate: historical and projected paths for the Euro Area*, Journal of Economic Dynamics and Control, v. 132, Article 04209, WP 1306 (November 2020).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility*, Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018)**.
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021)**.
- FAIELLA I. and A. MISTRETTA, The net zero challenge for firms' competitiveness, Environmental & Resource Economics, v. 83, pp. 85-113, WP 1259 (February 2020).
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, v.143, 3, pp. 1209-1226, **WP 1252 (December 2019).**
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).
- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference,* Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, **WP 1235 (October 2019).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).

- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, WP 1189 (September 2018).
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions*, Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

FORTHCOMING

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of text-based indicators in forecasting Italian economic activity*, International Journal of Forecasting, WP 1321 (March 2021).
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, **WP 1326 (March 2021).**
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *It ain't where you're from, it's where you're at: hiring origins, firm heterogeneity, and Wages, Journal of Econometrics,* WP 1374 (June 2022).
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, **WP 1358 (February 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, WP 1336 (June 2021).
- LOBERTO M, Foreclosures and house prices, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, **WP 1171 (April 2018).**
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy*, The Journal of Law and Economics, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, WP 1312 (December 2020).